1	On the use of inhibitors of 4-hydroxyphenylpyruvate
2	dioxygenase as a vector-selective insecticide in the
3	control of mosquitoes
4	
5	Marlon A. V. Ramirez <sup>1</sup> , Marcos Sterkel <sup>2¶*</sup> , Ademir de Jesus Martins <sup>3,4,5</sup> , José Bento
6	Pereira Lima <sup>3,4</sup> and Pedro L. Oliveira <sup>1,4</sup> ¶*
7	
8	<sup>1</sup> Laboratório de Bioquímica de Artrópodes Hematófagos, Instituto de Bioquímica Médica
9	Leopoldo de Meis, Universidade Federal do Rio de Janeiro, Rio de Janeiro, RJ, Brazil.
10	<sup>2</sup> Centro Regional de Estudios Genómicos, Universidad Nacional de La Plata (CREG-
11	UNLP), Argentina.
12	<sup>3</sup> Laboratorio de Fisiologia e Controle de Artrópodes Vetores, Instituto Oswaldo Cruz,
13	FIOCRUZ, Rio de Janeiro, RJ, Brasil.
14	<sup>4</sup> Laboratório de Entomologia, Instituto de Biologia do Exército, Rio de Janeiro, RJ, Brasil.
15	<sup>5</sup> Instituto Nacional de Ciencia e Tecnologia em Entomologia Molecular (INCT-EM),
16	Brazil.
17	
18	* Corresponding author
19	E-mail addresses:
20	pedro@bioqmed.ufrj.br
21	msterkel@conicet.gov.ar

<sup>22</sup> <sup>¶</sup> These authors contributed equally to this work.

23

24

# 25 Abstract

26 Blood-sucking insects incorporate many times their body weight of blood in a single meal. As proteins are the major component of vertebrate blood, its digestion in the 27 gut of hematophagous insects generates extremely high concentrations of free amino 28 acids. Previous reports showed that the tyrosine degradation pathway plays an essential 29 30 role in adapting these animals to blood feeding. Inhibiting 4-hydroxyphenylpyruvate 31 dioxygenase (HPPD), the rate-limiting step of tyrosine degradation, results in the death of insects after a blood meal. Therefore, it was suggested that compounds that block the 32 33 catabolism of tyrosine could act selectively on blood-feeding insects. Here we have evaluated the toxicity against mosquitoes of three HPPD inhibitors currently used as 34 35 herbicides and in human health. Among the compounds tested, nitisinone (NTBC) proved to be more potent than mesotrione (MES) and isoxaflutole (IFT) in Aedes aegypti. 36 37 NTBC was lethal to Ae. aegypti in artificial feeding assays (LD50: 4.36 µM), as well as in topical application (LD50: 0.0033 nmol/mosquito). NTBC was also lethal to Ae. aegypti 38 populations that were resistant to neurotoxic insecticides, and it was lethal to other 39 40 mosquito species (Anopheles and Culex). Therefore, HPPD inhibitors, particularly NTBC, represent promising new drugs for mosquito control. Since they only affect blood-41 42 feeding organisms, they would represent a safer and more environmentally friendly alternative to conventional neurotoxic insecticides. 43

44

45

# 47 Author Summary

48 The control of mosquitoes has been pursued in the last decades by the use of neurotoxic insecticides to prevent the spreading of dengue, zika and malaria, among 49 other diseases. However, the selection and propagation of different mechanisms of 50 resistance hinder the success of these compounds. New methodologies are needed for 51 52 their control. Hematophagous arthropods, including mosquitoes, ingest quantities of blood that represent many times their body weight in a single meal, releasing huge 53 amounts of amino acids during digestion. Recent studies showed that inhibition of the 54 tyrosine catabolism pathway could be a new selective target for vector control. Thus we 55 tested three different inhibitors of the second enzyme in the tyrosine degradation 56 57 pathway as tools for mosquito control. Results showed that Nitisinone (NTBC), an inhibitor used in medicine, was the most potent of them. NTBC was lethal to Aedes 58 aegypti when it was administered together with the blood meal and when it was topically 59 applied. It also caused the death of Anopheles aguasalis and Culex guinguefasciatus 60 mosquitoes, as well as field-collected Aedes populations resistant to neurotoxic 61 62 insecticides, indicating that there is no cross-resistance. We discuss the possible use of 63 NTBC as a new insecticide.

64

65

## 66 Introduction

Mosquitoes are important vectors for pathogens that cause diseases such as malaria, lymphatic filariasis, yellow fever, dengue, chikungunya, Zika and West Nile fever. Mosquito-borne diseases are among the leading global public health menaces [1], and vector control by means of insecticides is crucial for the management of these diseases [2,3]. Conventional insecticides are applied in the stage of larva and adult. To

reduce these immature forms in the case of Ae. aegypti, temephos (organophosphate) 72 has been used for many years, as well as mechanical control to eliminate standing water 73 74 [4]. For adult control, malathion (organophosphate) and pyrethroids are mainly 75 recommended by World Health Organization (WHO). In the case of Anopheles 76 mosquitoes, insecticide-treated nets (ITNs) and indoor residual spraying (IRS) are used 77 as preventive strategies [5]. The main classes of insecticides are organochlorines, organophosphates, carbamates, and pyrethroids, all of them neurotoxic [6]. However, 78 79 their extensive use has led to the development of resistance to these insecticides [7,8], 80 representing a problem for mosquito control. This is particularly true for arbovirus transmitted by Ae. aegypti, exemplified by the recent global Zika outbreak [9], making 81 the search for alternative methods for mosquito control a high priority of the global public 82 83 health agenda [10].

84 Female mosquitoes need to feed on blood for the maturation of their eqgs. Since 85% of vertebrate blood dry weight is protein, its digestion generates high concentrations 85 of amino acids in the gut [11]. Although amino acids are considered essential nutrients, 86 87 many human genetic diseases are caused by defective degradation of amino acids, leading to hyperaminoacidemias and the formation of toxic metabolites [12]. In the 88 hematophagous "kissing bug" Rhodnius prolixus, a high level of expression of enzymes 89 related to tyrosine degradation is found in the midgut [13]. Silencing of 4-90 91 hydroxyphenylpyruvate dioxygenase (HPPD), the enzyme that catalyzes the second 92 (and rate-limiting) step of the tyrosine degradation pathway, led to insect death after a blood meal. Chemical inhibitors of HPPD caused the death of hematophagous 93 arthropods (kissing bug, mosquitoes and ticks) but were not toxic to non-hematophagous 94 95 insects [14]. Therefore, the degradation of free tyrosine formed in excess during 96 digestion of a blood meal is an essential trait in the adaptation to a hematophagous way 97 of life [11]. The key role of this pathway in the evolution of blood-feeding organisms led

us to identify tyrosine degradation as a potential target in the development of novel
alternative insecticides that would therefore be selective for these animals.

100 In plants, functional HPPD is required for the synthesis of plastoquinone and 101 tocopherol, which are essential for the plant to survive. Therefore, HPPD has been 102 identified as one of the most promising targets for the development of new herbicides, 103 and thousands of HPPD inhibitors have been synthesized [15]. HPPD inhibitors are 104 classified into three main chemical families: triketones, diketonitriles and pyrazolones 105 [16,17]. Triketones can be natural compounds such as leptospermone, or synthetic, such 106 as mesotrione (MES), nitisinone (NTBC) and many others [18]. MES is used as a 107 herbicide (Callisto®, Syngenta) [19]. NTBC, on the other hand, is approved for 108 therapeutic use in humans to treat hereditary tyrosinemia type 1 (HT-1) since 1994 109 (Orfadin®) [20] and its potential for use in patients with alkaptonuria is under investigation 110 [21,22]. In the diketonitriles group, isoxaflutole (IFT) is also used as a herbicide 111 (Balance® and Merlin®, Bayer) [23]. To test the possibility of using HPPD inhibitors as selective insecticides for disease vectors, here we investigated the toxicity of three 112 inhibitors of HPPD (NTBC, MES and IFT) towards mosquitoes, especially against Ae. 113 aegypti, evaluating different doses and possible modes of application. We also tested 114 115 NTBC toxicity to Ae. aegypti populations resistant to pyrethroids and organophosphates, as well as other species of mosquitoes, Culex guinguefasciatus and Anopheles 116 117 aquasalis. Our results support the use of HPPD inhibitors, particularly NTBC, as new 118 insecticides for mosquito control.

119

120

121

122

# 123 Methods

124

### 125 Ethics statement

All experiments were conducted according to the guidelines of the institutional care and use committee (Committee for Evaluation of Animal Use for Research from the Federal University of Rio de Janeiro, CAUAP-UFRJ), which is based on the NIH Guide for the Care and Use of Laboratory Animals (ISBN 0-309-05377-3). The protocols used here were approved by CAUAP-UFRJ under registry #IBQM155/13.

### 131 Mosquito rearing

Ae. aegypti Red Eye strain were maintained in the insectary of the Federal 132 University of Rio de Janeiro (UFRJ), Brazil. The insecticide-resistant Ae. aegypti 133 populations, originally collected with ovitraps from different Brazilian cities: Santarém 134 135 (Pará state), Nova Iguaçu (Rio de Janeiro state), and Oiapoque (Amapá state), were reared in the insectary at FIOCRUZ-RJ. A Rock-kdr strain (or R2R2 strain) that presents 136 137 point mutations in the voltage-dependent sodium channel associated with pyrethroid 138 resistance was also maintained in the FIOCRUZ-RJ insectary [24,25]. The Ae. aegypti 139 Rockefeller strain (also maintained at FIOCRUZ-RJ), which is commonly used as a standard for insecticide susceptibility assays, was used here as a reference strain, in 140 order to allow better comparison with literature data [26]. RR50 (Resistance rate 50: 141 LD50 of strain or population studied/LD50 of Rockefeller strain) for deltamethrin was: 142 143 Santarém population = 30.4, Nova Iguacu population = 25.4, Oiapogue population = 143.9 [27-30]. In the case of Rock-kdr, a knockdown time assay with deltamethrin 144 revealed that the time necessary to knockdown 95% of the Rock-kdr lineage was 6.7x 145 longer than the time found for the Rockefeller susceptible strain [25]. 146

147 The larvae were fed with powdered cat food (Friskies<sup>®</sup>, Nestlé Purina PetCare). 148 Adult mosquitoes were kept in cages and fed with a 10% sucrose solution. All 149 mosquitoes were reared at 26°C, in 70-80% relative humidity and a photoperiod of 12h 150 light:12h dark.

A colony of *An. aquasalis* was established in 1995 using specimens collected in the municipality of Guapimirim, Rio de Janeiro, and reared in the insectary of FIOCRUZ-RJ. The larvae were reared on a diet of fish food (Tetra Marine Large Flakes, Tetra GmH) in containers containing dechlorinated water at a concentration of 0.2 % NaCl (w/v). Adult mosquitoes were provided with 10 % sucrose *ad libitum* under a regimen of photoperiod, temperature and humidity similar to that of *Ae. aegypti* [31].

157 *Cx. quinquefasciatus* were reared at Instituto de Biologia do Exército (IBEX), Rio 158 de Janeiro. The larvae were fed cat food (Friskies®, Nestlé Purina PetCare). To 159 encourage copulation, adults were housed in a dark room, because this mosquito feeds 160 on blood at night. Temperature and humidity were similar to the other mosquitoes.

161

### 162 **Topical application assays**

NTBC, MES or IFT (Sigma Chemical Co.) were diluted in acetone just before each experiment. After feeding on blood, mosquitoes were cold-anesthetized and placed in a glass petri dish on ice. Then, a volume of 0.5 μl of the HPPD inhibitor solution was topically applied with a micropipette on the abdomen of the insect. Survival was evaluated every 24 h for a week. Controls received only acetone.

168

169

## 171 Artificial feeding assays

172 Rabbit blood was collected with a sterile syringe containing heparin at a ratio of 173 1 µl heparin stock/ml blood (heparin stock was 5000 IU/ml). Stock solutions of HPPD 174 inhibitors in PBS (NaCl 0.15 M, Na phosphate 10 mM, pH 7.0) were diluted in PBS and 175 mixed 1:9 (v/v) with heparinized blood to obtain final concentrations used to feed mosquitoes, as indicated in figure legends. Controls received PBS in blood (1:9, v/v). 176 Mosquitoes of 3-5 days post-emergence were fed in an artificial feeding apparatus where 177 178 food was offered through a membrane of Parafilm M®. The temperature of the blood meal was maintained with a circulating water bath, adjusted at 37-38°C [32]. The 179 180 maximum feeding time was 30 minutes. Only fully engorged mosquitoes were used. 181 Survival was evaluated every 24 h for a week.

182

#### 183 Survival experiments, statistical analysis

Mosquitoes were offered a 10% sucrose solution ad libitum and were considered 184 dead if they could no longer stand. Statistical analysis and design of graphs were 185 186 performed using Prism 6.0 software (GraphPad Software, San Diego, CA). At least two 187 independent experiments were performed for each experimental condition (each with its respective control group). The Kaplan-Meier survival curve analysis in the Prism software 188 189 (the log rank test) was used to evaluate significant differences between experimental and control groups. LD50 was determined using a non-linear regression to fit 190 191 log[inhibitor] vs. normalized response (Variable slope). Two-way analysis of variance 192 and Tukey's multiple comparisons test were carried out to compare the field populations 193 and laboratory strains with Rockefeller strain.

194

# 196 **Results**

197 We examined three chemical inhibitors of HPPD for their effects on the survival of Ae. aegypti (Red Eye strain): two of them are marketed as herbicides (MES and IFT), 198 and one is used in medicine for the treatment of tyrosinemia type I (NTBC). MES and 199 200 NTBC are triketones while IFT belongs to the diketonitrile family. Inhibitors from the 201 family of pyrazoles were not evaluated. The three inhibitors were administered by 202 artificial feeding and topical application. NTBC was the most potent of the three inhibitors 203 in the artificial feeding trials, presenting an LD50 (the dose that kills 50% of mosquitoes) 204 of 4.36 µM, while MES presented an LD50 of 324 µM (Fig. 1, Table 1). IFT showed no 205 lethal effects in any of the concentrations tested. Thus, NTBC was about 74 times more potent than MES when it was co-administered along with the blood meal. 206

207

208

Figure 1. Ingestion of HPPD inhibitors with the blood meal decreases the survival 209 of Ae. aegypti (Red Eye strain). (A) Tyrosine catabolism pathway. TAT: tyrosine 210 aminotransferase; HPPD: 4-hydroxyphenylpyruvate dioxygenase; HgD: homogentisate 211 1, 2 dioxygenase; MAAI: maleylacetoacetate isomerase; FAH: fumarylacetoacetase. (B) 212 213 Survival rates of Ae. aegypti (Red Eye strain) fed with rabbit blood supplemented with 214 NTBC. PBM: Post-blood meal. Control group was fed with blood plus PBS (9:1; v/v). (C) 215 Survival rates of Ae. aegypti (Red Eye strain) fed with rabbit blood supplemented with 216 IFT (D) Dose-response curves at 72 h PBM. MES data were taken from Sterkel et al. (2016). Four (panel B) and two (panel C) independent experiments were performed 217 218 respectively, each with n =10-36 insects per experimental group. Panels B and C are plotted as Kaplan-Meier survival curves. 219

220

222

223

NTBC proved to be more potent than MES and IFT also in the topical application assay (Figure 2), presenting an LD50 = 0.0033 nmol/insect (1.1 ng/insect). However, in contrast to the artificial feeding assay, IFT also was lethal, presenting an LD50 = 20.4 nmol/insect (7320 ng/insect) (Table 1). These results demonstrated that NTBC was around 6182 times more potent than IFT. On the other hand, MES did not cause mortality in mosquitoes when applied topically, suggesting that this compound was not able to traverse the cuticle.

231

232

Figure 2. Topical application of HPPD inhibitors effect on survival of Ae. aegypti 233 234 (Red Eve strain) after a blood meal. Topical application of (A) NTBC or (B) IFT on the 235 abdomen causes mosquitoes death. (C) MES had very little effect. (D) Dose-response 236 curves recorded at 72 h PBM. Panels A, B and C are plotted as Kaplan-Meier survival 237 curves. Three (A) and two (B, C) independent experiments were performed, respectively, 238 each with n =12-35 insects per experimental group. The drugs (dissolved in 0.5 µl 239 acetone) were applied on the abdomen immediately after the blood meal (time 0 PBM). 240 Controls received only acetone (0.5 µl).

241

Table 1 – Toxicity of HPPD inhibitors to *Ae. aegypti* (Red Eye strain) using artificial feeding and topical application - LD50 were calculated from data in Figures 1 and 2. Data shown are mean  $\pm$  95% Confidence Interval (CI).

Inhibitor	LD50	(95% CI)
	Artificial feeding	Topical application

	(µM)	(nmol/insect)
NTBC	4.36 (3.85-4.59)	0.0033 (0.0019-0.006)
IFT	-	20.4 (17.70-23.43)
MES	324 (193.4-543.9)	-

245

246

The effect of NTBC on other mosquito species was evaluated in trials of artificial feeding. *Cx. quinquefasciatus* and *An. aquasalis* died when fed with blood containing concentrations of NTBC similar to those that killed *Ae. aegypti* (Red Eye and Rockefeller strains) (Fig. 3 and Table 2). These results show that NTBC can be used not only for the control of *Aedes* populations, but also for the control of other mosquitoes that transmit pathogens.

253

## Figure 3. Ingestion of NTBC is lethal to other mosquito species.

(A) *Ae. aegypti* (Rockefeller strain) were fed on rabbit blood plus NTBC. (B) *Cx. quinquefasciatus* were fed rabbit blood plus NTBC. (C) *An. aquasalis* were fed rabbit blood plus NTBC. (D) Dose-response curves at 72 h PBM. Panels A, B and C are plotted as Kaplan-Meier survival curves. 3 independent experiments were performed, each with n =10–37 insects per experimental group.

260

261

Table 2 – Toxicity of NTBC to different mosquito species using artificial feeding - LD50
were calculated from data in Figure 3. Data shown are mean ± 95% Confidence Interval
(CI).

Species of mosquito	<b>NTBC LD50 (</b> 95% CI)	
	μΜ	µg/ml
<i>Ae. aegypti</i> (Rockefeller strain)	1.94 (1.42-2.65)	0.64 µg/ml
Cx. quinquefasciatus	2.28 (1.56-3.34)	0.75 µg/ml
An. aquasalis	1.41 (1.03-1.93)	0.46 µg/ml

265

266 Next, we studied the effect of NTBC on field-collected populations of Ae. aegypti 267 that showed high levels of resistance to organophosphates and pyrethroids. NTBC was 268 also tested on the laboratory Rock-kdr strain, which carries a mutation in the voltagedependent sodium channel, the target site of pyrethroids and organochlorines (DDT). 269 270 When these populations were fed with NTBC-supplemented blood, they presented LD50 271 values similar to the control (Rockefeller strain) (Fig. 4 and Table 3). Thus neither field-272 collected Ae. aegypti populations nor the Rock-kdr strain showed cross-resistance to 273 NTBC when orally administered along with a blood meal.

274 Finally, topical application trials were also carried out on the insecticide-resistant 275 mosquitoes. In this case, unlike artificial feeding, there were significant differences 276 between field-collected neurotoxic-resistant populations and controls (Rockefeller strain) 277 (p<0.001). These populations were less susceptible to NTBC than controls, presenting 278 RR50 values of 5.1, 5.5 and 3.9 for Santarém, Nova Iquacu and Oiapoque populations, 279 respectively. In contrast, there were no significant differences between the LD50 values 280 calculated for Red Eye, Rock-kdr and Rockefeller strains (Figure 5 and Table 3). These 281 results might be explained by a reduced penetration of this drug through the cuticle, since no major differences were observed in NTBC LD50 during artificial feeding trials. 282

283

ົ	o	_ /
/	n	4
_	-	

285

Figure 4. Insecticide-resistant *Ae. aegypti* populations do not show crossresistance towards NTBC ingested with a blood meal. Different *Ae. aegypti* populations were fed with blood plus NTBC. (A) Santarém population, (B) Nova Iguaçu population, (C) Oiapoque population, (D) Rock-kdr strain. (E) Dose-response curves at 72 h PBM. Panels A-D are plotted as Kaplan-Meier survival curves. Three (A, B, C) and two (D) independent experiments were performed, respectively, each with n =10–35 insects per experimental group.

293

294

295

Figure 5. Insecticide-resistant Ae. aegypti populations show moderate resistance 296 297 to NTBC applied topically after a blood meal. Topical application of NTBC caused 298 death to Ae. aegypti (A) Rockefeller, (B) Santarém, (C) Nova Iguaçu, (D) Oiapoque; and 299 (E) Rock-kdr strain. (F) Dose-response curves at 72 h PBM. Panels A, B, C, D and E are 300 plotted as Kaplan-Meier survival curves. Four independent experiments for Rockefeller 301 strain and Oiapoque population and two independent experiments for the other mosquito 302 populations were performed, each with n = 10-32 insects per experimental group. The 303 drug (in 0.5 µl acetone) was applied on the abdomen immediately after a blood meal. 304 Controls received 0.5 µl acetone. Differences in the LD50 between Santarém, Nova 305 Iguaçu and Oiapoque populations (p<0.001) and Rockefeller strain were observed.

306

308 Table 3 – Toxicity of NTBC to insecticide-resistant Ae. aegypti populations - LD50 were

309 calculated from data in Figures 4 and 5. \*\*\*= p<0.001. Data shown are mean  $\pm$  95%

310 Confidence Interval (CI).

Ae. aegypti strains or	<b>NTBC LD50 (</b> 95% CI)	
populations	Artificial feeding	Topical application
	(µM)	(nmol/insect)
Rockefeller	1.94 (1.42-2.65)	0.0046 (0.0036-0.0060)
Santarém	2.82 (1.79-4.41)	0.0235 (0.017-0.03)***
Nova Iguaçu	3.20 (1.76-5.79)	0.0255 (0.018-0.037)***
Oiapoque	2.33 (1.94-2.76)	0.0181 (0.012-0.026)***
Rock-kdr	1.56 (1.39-1.75)	0.0075 (0.0052-0.011)

311

312 Toxicity of HPPD inhibitors toward hematophagous arthropods depends on the digestion of blood meal proteins, as previous data showed that sugar-fed female 313 314 mosquitoes were not sensitive to mesotrione [14]. Using topical/contact applications of 315 the drug under field conditions ensures that drug exposure does not occur at the same time as blood intake. Therefore, we decided to determine the maximum time interval 316 317 between drug exposure and blood feeding that would still ensure lethality of the drug. For this, mosquitoes were fed on blood at different times (0 h, 24 h, 48 h, 72 h, 96 h) 318 319 after the topical application of an LD95 of NTBC (15.6 ng/mosquito, the dose that kills 95% of the mosquitoes when it is applied immediately after a blood meal). Although a 320 321 significant toxicity was retained for the first 24 h after drug application, the longer the time 322 between blood feeding and topical application of NTBC, the lower the number of dead 323 mosquitoes, suggesting progressive inactivation and/or excretion of the drug by the

mosquito (Figure 6). In a second set of tests, mosquitoes were first fed on blood and then the topical application was made at different times PBM (0 h, 24 h, 48 h, 72 h) (Figure 7). Here, a similar time-dependent loss of efficacy is observed because by 48 h and later time points after the meal, there is little blood protein in the gut, and tyrosine has already been catabolized.

329

Figure 6. Blood meal administered at different times after topical application of 330 331 **NTBC.** To evaluate persistence of toxicity from topically applied NTBC on Ae. Aegypti 332 (Red Eye strain) the drug was applied at time 0 and the blood meal was administered at different times (dotted line): (A) 0 h, (B) 24 h, (C) 48 h, (D) 72 h, (E) 96 h. The results of 333 A-E are summarized as % mortality observed at 72 h PBM in (F). Panels A-E are plotted 334 as Kaplan-Meier survival curves. Two independent experiments were performed, each 335 with n =10–32 insects per experimental group. The LD95 (15.6 ng in 0.5  $\mu$ l acetone) was 336 337 applied on the abdomen of the mosquito at time 0. Acetone alone was applied to the 338 control group at 0 h PBM.

339

Figure 7. Topical application of NTBC at different times after PBM. Mosquitoes (Red 340 341 Eye strain) were fed with blood meal at time 0 and topical application of NTBC was 342 performed at different times (dotted line): (A) 0 h, (B) 24 h, (C) 48 h, (D) 72 h. The results of A-D are summarized as % mortality observed at 72 h after topical application (E). Data 343 344 from panels A to D were used to summarize the effect of time interval between NTBC 345 application and blood meal on mortality observed at 72 h after topical application. Data 346 in A-D are plotted as Kaplan-Meier survival curve. Two independent experiments were performed, each with n =10–32 insects per experimental group. 347

348

# 349 **Discussion and Conclusions**

Hematophagy in arthropods is linked to a hyperproteic diet to an extent not found 350 351 in other animals [11]. When the proteins in the blood are degraded, high levels of amino acids such as tyrosine accumulate in the digestive tract. The discovery that the capacity 352 353 to degrade free tyrosine produced in excess during blood meal digestion is an essential 354 trait in the physiology of blood-sucking arthropods that contributes to adapt these animals 355 to hematophagy led us to propose the use of HPPD inhibitors as a new class of 356 insecticides, selective for hematophagous animals [14]. In this study we evaluated the potential use of HPPD as a novel target for the control of mosquitoes, comparing different 357 358 inhibitors and modes of application. Our results showed that neither MES nor IFT was 359 very powerful in Ae. aegypti (Red Eye strain), while NTBC stood out for its potency and efficacy. This was consistent with the report of Sterkel et al. (2016), who found that 360 feeding of Ae. aegypti with rabbit blood supplemented with MES decreased their survival, 361 as did feeding them on mice treated with an orally applied therapeutic dose of NTBC 362 [14]. However, differences in the results depending on the mode of administration 363 364 provided relevant insights that can help further research on the use of HPPD inhibitors 365 as vector-selective insecticides. NTBC and MES, but not IFT, were effective against Ae. 366 aegypti when administered along with the blood meal (Figure 1), but topical application resulted in a different profile. Although much higher doses were required, NTBC and IFT 367 368 applied topically, but not MES, were effective.

An open question concerns the molecular mechanism responsible for the differences between MES, IFT and NTBC in response to topical application and artificial feeding, knowing that these compounds act on the same enzyme, HPPD. Whereas MES and NTBC act directly on the HPPD enzyme[19,33], IFT undergoes a biotransformation to the diketonitrile derivative (DKN), and it is this compound that acts on the HPPD enzyme [23,34]. Specifically, the difference between MES and IFT in these two assays might be explained either by differential absorption or by differential metabolic

376 modification of these compounds when administered by each route (cuticle and midgut). 377 Thousands of compounds are presently listed as HPPD inhibitors [15]; therefore, it is 378 important to highlight that this differential toxicity depending on the mode of 379 administration calls for a systematic comparison across a broad spectrum of inhibitors, 380 as a way to reveal alternative compounds directed to the same enzyme.

381 The spreading of insecticide resistance among natural populations is a factor that 382 has limited their efficiency in the control of vector-borne diseases and hence, fueled the 383 search for alternative methods. Since mosquitoes can produce many generations per year, resistance can evolve very quickly. Besides, the appearance of cross-resistance 384 385 among different neurotoxic compounds is a common finding [7,35,36]. A combination of 386 different mechanisms, such as metabolic resistance, mutation of the target proteins and 387 penetration factors (cuticular resistance) contribute to the resistance of insects to contact insecticides. Here we searched for cross-resistance using populations known to be 388 389 resistant to organophosphates and pyrethroids, where both metabolic resistance 390 (increased expression of detoxifying enzymes) and target-site mutations (kdr) are at play 391 [24,25,27-30]. No evidence for cross-resistance to NTBC appeared using oral 392 administration (Figure 4, Table 3), but when using the topical application assays a 393 moderate (3.9 to 5.5 fold) but significant (p<0.001) increase in the LD50 was observed 394 in Santarem, Nova Iguaçu and Oiapoque populations (Figure 5, Table 3). These results 395 might be explained by a lower penetration of NTBC through the cuticle as a possible 396 additional resistance mechanism present in these field populations, that complemented 397 the role of metabolic resistance and target-site mutations. Several studies have reported 398 increased expression of genes related to cuticle formation in resistant mosquito 399 populations [37,38]. However, this does not fully explain the high levels of resistance to 400 neurotoxic insecticides observed in these populations, pointing to a multifactorial nature 401 of resistance.

402 Rock-kdr strain is derived from the backcrossing of a field-derived population (homozygous for kdr mutations) and the Rockefeller strain for eight generations, to 403 404 reduce the contribution of detoxification enzymes (glutathione-S-transferase, esterases 405 and multifunction oxidases) in pyrethroid resistance and to evaluate the effect of the kdr 406 mutations alone [24]. As expected, since the molecular targets are different, when NTBC 407 was orally or topically administered to the Rock-kdr strain, there was no significant 408 variation in the LD50 with respect to the Rockefeller control strain. Similar results were 409 observed when comparing the Red Eye strain and the Rockefeller strain (Figure 5, Table 410 3). Furthermore, NTBC was also lethal for An. aquasalis and Cx. quinquefasciatus with potency similar to that observed in Ae. aegypti (Red Eye and Rockefeller strains), 411 412 reinforcing the hypothesis that it can be used to target several vector-borne diseases at 413 the same time.

The effectiveness of topical application of NTBC suggests that it can be used in 414 strategies such as indoor residual spraying (IRS) and Long-Lasting Insecticide-treated 415 416 Nets (LLINs). However, HPPD inhibitors have a lethal effect only in blood-fed insects, a 417 particular characteristic that makes them selective for hematophagous arthropods. This 418 fact also creates some limitations for their use in topical-application strategies (such as 419 LLIN and IRS), as illustrated by the results showing that the efficacy of NTBC in topical 420 application trials strongly depended on the time interval between drug administration and 421 blood meal intake. When mosquitoes were fed 24 hours after the topical application of an LD95 (the dose that killed 95% of mosquitoes when applied immediately after 422 423 feeding), it only killed 50% of the insects, indicating that a significant proportion (around 50%) of the drug had already been inactivated or excreted by that time. NTBC LD95 only 424 425 killed 25% of the mosquitoes when fed 72 hours after application, and it was not effective 426 when mosquitoes were fed later on (Fig. 6). When NTBC was applied at different times 427 after a blood meal (PBM), it was lethal when applied up to 48 hours PBM, indicating that most of the tyrosine had already been catabolized by that time (Fig. 7). Taken together, 428

our data demonstrate that NTBC may be useful as a lead compound for developing 429 compounds with a longer active life in mosquitoes. A wide range of other HPPD inhibitors 430 431 have already been identified in the search for herbicides and should be investigated as 432 possible tools for mosquito control. Additionally, the mortality determined by NTBC is not as fast as with the neurotoxic insecticides, and some fraction of a population of treated 433 434 females may survive longer than the gonotrophic cycle, which takes around 2-4 days 435 after the blood meal [39,40]. This fact may contribute to maintain susceptible alleles in 436 the population, slowing the evolution of resistance. This would be an effect similar to that of the idealized late-acting insecticides [41]. 437

438 The most used endectocide drug (it has activity against endo- and ectoparasites when applied to the host) in human and livestock is Ivermectin, capable of killing a wide 439 440 variety of parasites and vectors [42,43]. It targets a broad range of parasites and invertebrates, including mosquitoes [44-47], and it has been proposed as an additional 441 tool to control vector-borne diseases such as malaria [48-54]. However, ivermectin's 442 443 half-life in humans is short (around 18 h) and it would be necessary to administer multiple doses, a limitation in terms of logistics [52]. A recent study proposed the use of 444 445 isoxazoline drugs (fluralaner and afoxolaner), currently used against fleas and ticks 446 infesting animals, for the drug-based oral treatment of a proportion of human population for the control of Ae. aegypti, Culex pipiens, Anopheles and sand flies [55]. These 447 448 compounds possess long *in-vivo* half-lives that provide weeks to months of protection after a single oral administration. However, they are not approved for use in humans and 449 450 there is no information about their pharmacokinetic and/or possible side effects. In 451 contrast, NTBC has been used in humans for treatment of HT-1 since 1994. It is 452 remarkably safe drug for mammals (LD50>1000 mg/kg in rats) [56] presenting a half-life 453 of 54 h in human plasma. Its concentration in blood following the ingestion of a therapeutic dose (1 mg/kg) is 8 µg/ml (24.3 µM) [57], much higher than the LD95 454 455 observed for all mosquito species during artificial feeding experiments. Because NTBC

toxicity towards hematophagous arthropods is maximized when it is co-delivered along
with the blood meal, our results raise the possibility that it could be used as a new
endectocide drug (in humans and livestock) for control of mosquito-borne diseases, as
part of an integrated vector management program.

460

461

# 462 Acknowledgments

We wish to thank Dr. Martha Sorenson for a critical reading and for all suggestions in the writing of the manuscript. We thank all of the members of the Laboratório de Bioquímica de Artrópodes Hematófagos (UFRJ), especially JM Freire for breeding of *Ae. aegypti* (Red Eye strain) and J Margues, C Cosme and SR Cássia for technical assistance.

We also thank members of the Laboratório de Fisiologia e Controle de Artrópodes Vetores (FIOCRUZ), especially L dos Santos Dias, for suggestions for conducting the trials and for providing the resistant populations from Santarém and Nova Iguaçu, and L Carrara for providing Oiapoque population and Rock-kdr strain. We thank R Santos, P Serravale and Q Amorim for the breeding of *An. aquasalis* and *Cx. quinquefasciatus*.

472

# 473 **References:**

- 474 1. Marshall E. A renewed assault on an old and deadly foe. Science.
  475 2000;290: 428–30. doi:10.1126/science.290.5491.428
- Bhatt S, Weiss DJ, Cameron E, Bisanzio D, Mappin B, Dalrymple U, et al.
   The effect of malaria control on Plasmodium falciparum in Africa between
   2000 and 2015. Nature. 2015; doi:10.1038/nature15535

- 479 3. Liu N. Insecticide resistance in mosquitoes: Impact, mechanisms, and
  480 research directions. Annu Rev Entomol. 2015;60: 537–559.
  481 doi:10.1146/annurev-ento-010814-020828
- George L, Lenhart A, Toledo J, Lazaro A, Han WW. Community Effectiveness of Temephos for Dengue Vector Control: A Systematic
   Literature Review. PLoS Negl Trop Dis. 2015;
   doi:10.1371/journal.pntd.0004006
- WHO. Test procedures for insecticide resistance monitoring in malaria
  vector mosquitoes. Second. World Health Organisation Technical Report
  Series. 2016. doi:10.1007/978-3-642-10565-4
- 6. Costa LG, Giordano G, Guizzetti M, Vitalone A. Neurotoxicity of pesticides:
  a brief review. Front Biosci. 2008;13: 1240–1249. doi:10.2741/2758
- Moyes CL, Vontas J, Martins AJ, Ng LC, Koou SY, Dusfour I, et al.
  Contemporary status of insecticide resistance in the major Aedes vectors
  of arboviruses infecting humans. PLoS Negl Trop Dis. 2017; 1–20. doi:
  10.1371/journal.pntd.0005625
- 8. Ranson H, Lissenden N. Insecticide resistance in african anopheles
  mosquitoes : A worsening situation that needs urgent action to maintain
  malaria control. Trends Parasitol. Elsevier Ltd; 2015;xx: 1–10.
  doi:10.1016/j.pt.2015.11.010
- Messina JP, Kraemer MUG, Brady OJ, Pigott DM, Shearer FM, Weiss DJ,
  et al. Mapping global environmental suitability for Zika virus. Elife. 2016;5:
  1–19. doi:10.7554/eLife.15272

502	10.	Boëte C, Reeves RG. Alternative vector control methods to manage the
503		Zika virus outbreak: More haste, less speed. Lancet Glob Heal. 2016;4:
504		e363. doi:10.1016/S2214-109X(16)00084-X

- 505 11. Sterkel M, Oliveira JHM, Bottino-Rojas V, Paiva-Silva GO, Oliveira PL. The
  506 dose makes the poison: Nutritional overload determines the life traits of
  507 blood-feeding arthropods. Trends Parasitol. Elsevier Ltd; 2017;33: 633–
  508 644. doi:10.1016/j.pt.2017.04.008
- Scott CR. The genetic tyrosinemias. Am J Med Genet Part C Semin Med
   Genet. 2006;142C: 121–126. doi:10.1002/ajmg.c.30092
- 13. Ribeiro MC, Genta FA, Sorgine MHF, Logullo R, Mesquita RD, Paiva-silva
  GO, et al. An insight into the transcriptome of the gigestive tract of the
  bloodsucking bug , Rhodnius prolixus. PLoS Negl Trop Dis. 2014;8: 1–31.
  doi:10.1371/journal.pntd.0002594
- 515 14. Sterkel M, Perdomo HD, Guizzo MG, Barletta ABF, Nunes RD, Dias FA, et
  516 al. Tyrosine detoxification is an essential trait in the life history of blood517 feeding arthropods. Curr Biol. Elsevier Ltd.; 2016;26: 2188–2193.
  518 doi:10.1016/j.cub.2016.06.025
- 15. Beaudegnies R, Edmunds AJF, Fraser TEM, Hall RG, Hawkes TR, Mitchell 519 G, et al. Herbicidal 4-hydroxyphenylpyruvate dioxygenase inhibitors-A 520 review of the triketone chemistry story from a Syngenta perspective. 521 Bioorganic Med Elsevier 2009;17: 522 Chem. Ltd; 4134-4152. doi:10.1016/j.bmc.2009.03.015 523
- 16. Hirai K, Uchida A, Ohno R. Major synthetic routes for modern herbicide

525	classes and agrochemical characteristics. In: Böger P, Wakabayashi K,
526	Hirai K, editors. Herbicides classes in development. Srpinger; 2002. pp.
527	221–229. doi:10.1007/978-3-642-59416-8

- 17. Van Almsick A. New HPPD-inhibitors A proven mode of action as a new
   hope to solve current weed problems. Outlooks Pest Manag. 2009;20: 27–
- 530 30. doi:10.1564/20feb09
- 18. Knudsen CG, Lee DL, Michaely WJ, Chin H-L, Nguyen N, Rusay RJ, et al.
  Discovery of the triketone class of HPPD inhibiting herbicides and their
  relationship to naturally occurring β-triketones. In: S.S.Narwall, E.Hoagland
  R, R.H.Dilday, M.J.Reigosa, editors. Allelopathy in Ecological Agriculture
  and Forestry. Springer; 2000. pp. 101–111. doi:10.1007/978-94-011-41734
- Mitchell G, Bartlett DW, Fraser TEM, Hawkes TR, Holt DC, Townson JK,
  et al. Mesotrione: A new selective herbicide for use in maize. Pest Manag
  Sci. 2001;57: 120–128. doi:10.1002/1526-4998(200102)57:2<120::AID-</li>
  PS254>3.0.CO;2-E

Lindstedt, Holme E, Lock EA, Hjalmarson O SB. Treatment of hereditary
tyrosinaemia type I by inhibition of 4-hydroxyphenylpyruvate dioxygenase.
Lancet. 1992; 813–817.doi: 10.1016/0140-6736(92)92685-9

Suwannarat P, O'Brien K, Perry MB, Sebring N, Bernardini I, Kaiser-Kupfer
MI, et al. Use of nitisinone in patients with alkaptonuria. Metabolism.
2005;54: 719–728. doi:10.1016/j.metabol.2004.12.017

547 22. Introne WJ, Perry MB, Troendle J, Tsilou E, Kayser MA, Suwannarat P, et

al. A 3-year randomized therapeutic trial of nitisinone in alkaptonuria. Mol
Genet Metab. Elsevier B.V.; 2011;103: 307–314.
doi:10.1016/j.ymgme.2011.04.016

Pallett KE, Cramp SM, Little JP, Veerasekaran P, Crudace AJ, Slater AE.
Isoxaflutole: The background to its discovery and the basis of its herbicidal
properties. Pest Manag Sci. 2001;57: 133–142. doi:10.1002/15264998(200102)57:2<133::AID-PS276>3.0.CO;2-0

24. Brito LP, Linss JGB, Lima-Camara TN, Belinato TA, Peixoto AA, Lima JBP, 555 et al. Assessing the effects of Aedes aegypti kdr mutations on pyrethroid 556 557 resistance and its fitness cost. PLoS One. 2013;8: 1–10. doi:10.1371/journal.pone.0060878 558

25. Brito LP, Carrara L, Freitas RM De, Bento J, Lima P, Martins AJ. Levels of 559 resistance to pyrethroid among distinct kdr alleles in Aedes aegypti 560 laboratory lines and frequency of kdr alleles in 27 natural populations from 561 562 Rio de Janeiro Brazil. Biomed Res Int. 2018;2018: 1–10. doi:10.1155/2018/2410819 563

Kuno G. Early history of laboratory breeding of Aedes aegypti (Diptera:
Culicidae) focusing on the origins and use of selected strains. BioOne.
2010;47: 957–971. doi:10.1603/ME10152

Dos Santos Dias L, MacOris MDLDG, Andrighetti MTMO, Otrera VCG,
Dias ADS, Bauzer LGSDR, et al. Toxicity of spinosad to temephos-resistant
Aedes aegypti populations in Brazil. PLoS One. 2017;12: 1–15.
doi:10.1371/journal.pone.0173689

571 28. Dos Santos Dias L. Avaliação da persistência e efeito do spinosad no
572 desenvolvimento e reprodução de populações brasileiras de Aedes aegypti
573 (Diptera: Culicidae) resistentes aos inseticidas temephos e deltametrina.
574 Dissertação. Fundação Oswaldo Cruz. 2015. p. 127f. Available from:
575 https://sucupira.capes.gov.br/sucupira/public/consultas/coleta/trabalhoCo
576 nclusao/viewTrabalhoConclusao.jsf?popup=true&id\_trabalho=2366813

29. Costa M de M. Avaliação da resistência a inseticidas e mecanismos 577 selecionados em populações de Aedes aegypti Linnaeus 1762 (Diptera, 578 Culicidae) da fronteira entre Brasil e Guiana Francesa. Dissertação. 579 Fundação Oswaldo Cruz. 2017. 123f. Available 580 p. from: https://www.arca.fiocruz.br/handle/icict/27958 581

582 30. Bottino-Rojas V, Talyuli OAC, Carrara L, Martins AJ, James AA, Oliveira PL, et al. The redox-sensing gene Nrf2 affects intestinal homeostasis, 583 584 insecticide resistance and Zika virus susceptibility in the mosquito Aedes aegypti. J Biol Chem. 2018;293: jbc.RA117.001589. 585 doi:10.1074/jbc.RA117.001589 586

31. Pimenta PFP, Orfano AS, Bahia AC, Duarte APM, Ríos-velásquez CM,
Melo FF, et al. An overview of malaria transmission from the perspective of
Amazon Anopheles vectors. Mem Inst Oswaldo Cruz. 2015;110: 23–47.
doi:10.1590/0074-02760140266

32. Rutledge L., Ward R., Gould D. Studies on the feeding response of
mosquitoes to nutritive solutions in a new membrane feeder. Mosq News.
1964;24: 407–19.

33. Ellis MK, Whitfield AC, Gowans LA, Auton TR, Provan WM, Lock EA, et al.

Inhibition of 4-Hydroxyphenylpyruvate dioxygenase by 2-(2-Nitro-4trifluoromethylbenzoyl)-cyclohexane-1,3-dione and 2-(2-Chloro-4methanesulfonylbenzoyl)-cyclohexane-1,3-dione. Toxicology and Applied
Pharmacology. 1995. pp. 12–19. doi:10.1006/taap.1995.1121

- Garcia I, Job D, Matringe M. Inhibition of p-Hydroxyphenylpyruvate
  dioxygenase by the diketonitrile of isoxaflutole: A case of half-site reactivity.
  Biochemistry. 2000;39: 7501–7507. doi:10.1021/bi000135h
- Flores AE, Ponce G, Silva BG, Gutierrez SM, Bobadilla C, Lopez B, et al.
  Wide spread cross resistance to pyrethroids in Aedes aegypti (Diptera:
  Culicidae) from Veracruz State Mexico. J Econ Entomol. 2013;106: 959–
  doi:10.1603/EC12284
- 36. Hemingway J, Ranson H. Insectide resistance in insect vectors of human
  disease. Annu Rev Entomol. 2000;45: 371–391.
  doi:10.1146/annurev.ento.45.1.371
- Seixas G, Grigoraki L, Weetman D, Vicente JL, Silva AC, Pinto J, et al.
  Insecticide resistance is mediated by multiple mechanisms in recently
  introduced Aedes aegypti from Madeira Island (Portugal ). PLoS Negl Trop
  Dis. 2017; 1–16. doi:10.1371/journal.pntd.0005799
- 38. David J, Coissac E, Melodelima C, Poupardin R, Riaz MA, Chandor-proust
  A, et al. Transcriptome response to pollutants and insecticides in the
  dengue vector Aedes aegypti using next-generation sequencing
  technology. BMC Genomics. 2010;11: 1–12. doi:10.1186/1471-2164-11216

618	39.	Charlwood JD, Smith T, Billingsley PF, Takken W, Lyimo EOK, Meuwissen
619		JHET. Survival and infection probabilities of anthropophagic anophelines
620		from an area of high prevalence of Plasmodium falciparum in humans. Bull
621		Entomol Res. 1997;87: 445–453. doi:10.1017/S0007485300041304
622	40.	Killeen G, McKenzie E, Foy B, Schieffelin C, Billingsley P, Beier J. A
623		simplified model for predicting malaria entomologic inoculation rates based
624		on entomologic and parasitologic parameters relevant to control. Am J Trop
625		Med Hyg. 2000;62: 535–544. doi:10.4269/ajtmh.2000.62.535
626	41.	Read AF, Lynch PA, Thomas MB. How to make evolution-proof insecticides
627		for malaria control. PLoS Biol. 2009;7: 1–10.
628		doi:10.1371/journal.pbio.1000058
629	42.	Crump A, Omura S. Review Ivermectin, 'Wonder drug' from Japan: the
630		human use perspective. Proc Japan Acad Ser B. 2011;87: 13-28.
631		doi:10.2183/pjab.87.13
632	43.	Omura S, Crump A. Ivermectin and malaria control. Malar J. BioMed
633		Central; 2017;16: 1–3. doi:10.1186/s12936-017-1825-9
634	44.	Ameen M, Arenas R, Villanueva-Reyes J, Ruiz-Esmenjaud J, Millar D,
635		Domínguez-Dueñas F, et al. Oral ivermectin for treatment of pediculosis
636		capitis. Pediatr Infect Dis J. 2010;29: 991–993.
637		doi:10.1097/INF.0b013e3181e63f5f
638	45.	Chosidow O, Giraudeau B, Cottrell J, Izri A, Hofmann R, Mann SG, et al.
639		Oral ivermectin versus malathion lotion for difficult-to-treat head lice. N Engl
640		J Med. 2010;362: 896–905. doi:10.1056/NEJMoa0905471

46. Smit MR, Ochomo EO, Aljayyoussi G, Kwambai TK, Abong'o BO, Chen T,
et al. Safety and mosquitocidal efficacy of high-dose ivermectin when coadministered with dihydroartemisinin-piperaquine in Kenyan adults with
uncomplicated malaria (IVERMAL): a randomised, double-blind, placebocontrolled trial. Lancet Infect Dis. Elsevier Ltd; 2018;18: 615–626.
doi:10.1016/S1473-3099(18)30163-4

- 47. Panahi Y, Poursaleh Z, Goldust M. The efficacy of topical and oral
  ivermectin in the treatment of human scabies. Ann Parasitol. 2015;61: 11–
  6. Available from: http://www.ncbi.nlm.nih.gov/pubmed/25911032
- 48. Pooda HS, Rayaisse JB, Hien DFDS, Lefèvre T, Yerbanga SR, Bengaly Z,
  et al. Administration of ivermectin to peridomestic cattle: A promising
  approach to target the residual transmission of human malaria. Malar J.
  BioMed Central; 2015;14: 1–12. doi:10.1186/s12936-015-1001-z
- Kobylinski KC, Deus KM, Butters MP, Hongyu T, Gray M, da Silva IM, et
  al. The effect of oral anthelmintics on the survivorship and re-feeding
  frequency of anthropophilic mosquito disease vectors. Acta Trop. Elsevier
  B.V.; 2010;116: 119–126. doi:10.1016/j.actatropica.2010.06.001
- 658 50. Kobylinski KC, Sylla M, Chapman PL, Sarr MD, Foy BD. Ivermectin mass 659 drug administration to humans disrupts malaria parasite transmission in Hyg. Senegalese 2011;85: 660 villages. Am J Trop Med 3–5. doi:10.4269/ajtmh.2011.11-0160 661
- 51. Sylla M, Gray M, Chapman PL, Sarr MD, Rasgon JL. Mass drug
  administration of ivermectin in south-eastern Senegal reduces the
  survivorship of wild-caught, blood fed malaria vectors. Malar J. 2010;9: 1–

### 665 10. doi:10.1186/1475-2875-9-365

- Foy BD, Kobylinski KC, Silva IM da, Rasgon JL, Sylla M. Endectocides for
  malaria control. Trends Parasitol. Elsevier Ltd; 2011;27: 423–428.
  doi:10.1016/j.pt.2011.05.007
- Alout H, Krajacich B, Meyers J, Grubaugh N, Brackney D, Kobylinski K, et
  al. Evaluation of ivermectin mass drug dministration for malaria
  transmission control across different West African environments. Malar J.
  2014;13: 417. doi:10.1186/1475-2875-13-417
- Ouédraogo AL, Bastiaens GJH, Tiono AB, Guelbéogo WM, Kobylinski KC, 673 54. Ouédraogo A, et al. Efficacy and safety of the mosquitocidal drug 674 ivermectin to prevent malaria transmission after treatment: A double-blind, 675 Clin randomized. clinical Infect Dis. 2015:60: 357-365. 676 trial. doi:10.1093/cid/ciu797 677
- 678 55. Miglianico M, Eldering M, Slater H, Ferguson N, Ambrose P, Lees RS. Repurposing isoxazoline veterinary drugs for control of vector-borne 679 diseases. Natl 2018;115: 680 human Proc Acad Sci. 1-7. doi:10.1073/pnas.1801338115 681
- 56. Lock E, Ranganath LR, Timmis O. The Role of Nitisinone in Tyrosine
  Pathway Disorders. Curr Rheumatol Rep. 2014;16: 1–8.
  doi:10.1007/s11926-014-0457-0
- 57. Hall MG, Wilks MF, McLean Provan W, Eksborg S, Lumholt B.
   Pharmacokinetics and pharmacodynamics of NTBC (2-(2-nitro-4 fluoromethylbenzoyl)-1,3-cyclohexanedione) and mesotrione, inhibitors of

- 4-hydroxyphenyl pyruvate dioxygenase (HPPD) following a single dose to
  healthy male volunteers. Br J Clin Pharmacol. 2001;52: 169–177.
  doi:10.1046/j.0306-5251.2001.01421.x
- 691
- 692









D

Ε







